**Author’s response to reviews**

**Title:** Screening with the double surprise question to predict deterioration and death: an explorative study

**Authors:**

Carel Veldhoven (carel.veldhoven@radboudumc.nl)

Nienke Nutma (nienke.nutma@radboudumc.nl)

Willemijn de Graaf (wdegraaf@huisartsenpraktijkbergendal.nl)

Henk Schers (henk.schers@radboudumc.nl)

Stans Verhagen (stans.verhagen@radboudumc.nl)

Kris Vissers (kris.vissers@radboudumc.nl)

Yvonne Engels (yvonne.engels@radboudumc.nl)

**Version:** 5 **Date:** 06 Dec 2019

**Author’s response to reviews:**

Dear editor,

Last Tuesday, the first author Carel Veldhoven, who also is the corresponding author, changed this corresponding authorship on the website, but something seems to have go wrong.

As you can see on the title page, Carel Veldhoven will be the corresponding author, also after publication.

We added the full names of the laws, including links to websites where they can be found.

We hope that we addressed all minor remarks,

Best wishes, also on behalf of the corresponding author Carel Veldhoven, and the other authors,

Yvonne Engels